

Received September 23, 2020, accepted October 8, 2020, date of publication October 14, 2020, date of current version October 28, 2020. *Digital Object Identifier* 10.1109/ACCESS.2020.3031019

# **OpenSHC: A Versatile Multilegged Robot Controller**

BENJAMIN TAM<sup>®1,2</sup>, (Student Member, IEEE), FLETCHER TALBOT<sup>1</sup>, (Member, IEEE), RYAN STEINDL<sup>1</sup>, (Member, IEEE), ALBERTO ELFES<sup>1</sup>, (Member, IEEE), AND NAVINDA KOTTEGE<sup>®1</sup>, (Senior Member, IEEE)

<sup>1</sup>Robotics and Autonomous Systems Group, CSIRO, Pullenvale, QLD 4069, Australia

<sup>2</sup>School of Information Technology and Electrical Engineering, The University of Queensland, Brisbane, QLD 4072, Australia

Corresponding author: Navinda Kottege (navinda.kottege@csiro.au)

This work was supported by the Commonwealth Scientific and Industrial Research Organisation (CSIRO).

**ABSTRACT** Multilegged robots have the ability to perform stable locomotion on relatively rough terrain. However, the complexity of legged robots over wheeled or tracked robots make them difficult to control. This paper presents OpenSHC (Open-source Syropod High-level Controller), a versatile high-level controller capable of generating gaits and poses for quasi-static multilegged robots, both simulated and with real hardware implementations. With full Robot Operating System (ROS) integration, the controller can be quickly deployed on robots with different actuators and sensor payloads. The flexibility of OpenSHC is demonstrated on the 30 degrees of freedom hexapod Bullet, analysing the energetic performance of various leg configurations, kinematic arrangements and gaits over different locomotion speeds. With OpenSHC being easily configured to different physical and locomotion specifications, a hardware-based parameter space search for optimal locomotion parameters is conducted. The experimental evaluation shows that the insectoid configuration offers lower power consumption across a range of step frequencies; with the insectoid configuration providing performance advantages at higher body velocities and increased stability at low step frequencies. OpenSHC is open-source and able to be configured for various number of joints and legs.

**INDEX TERMS** Legged locomotion, legged robots, robot control, mechanism design.

## I. INTRODUCTION

Legged robots have advantages compared to their wheeled and tracked counterparts when navigating in complex terrain. From their ability to traverse discontinuous terrain, climb over obstacles and disturb the terrain minimally, to probing the terrain and manipulating the environment without an additional arm [1], [2]. However, the trade-off of this versatility and mobility is the significant challenges in mechanical and control complexity [3]. The combination of both the robot platform design and the control algorithm determines the performance and effectiveness of the robot. To further research and development of legged robots, the authors present Open-SHC - Open-source Syropod High-level Controller - a versatile controller that is capable of generating statically stable gaits for multilegged robots.<sup>1</sup> It is the result of legged

<sup>1</sup>https://github.com/csiro-robotics/syropod\_highlevel\_controller

robot locomotion research conducted at CSIRO's Robotics and Autonomous Systems Group since 2011, with some of the robots running OpenSHC shown in Fig. 1. Using the Robot Operating System (ROS) framework for modularity and easy deployment, OpenSHC is designed to generate foot tip trajectories for a given gait sequence, step clearance, step frequency and input body velocity for many different legged robots with various leg configurations and degrees of freedom. Any legged robot with up to 8 legs that can be specified using Denavit-Hartenberg (DH) parameters [4] can be used with OpenSHC. Input sensors such as IMUs and joint effort feedback can be utilised by the controller to provide robust trajectories in inclined and uneven terrain. OpenSHC can also be used in simulation to control robots in rviz and Gazebo simulation environments. This provides a convenient way for roboticists to design legged robots by being able to tune various parameters to fit specific performance criteria.

In order to show the capability of OpenSHC, a study on locomotion efficiency of a 30 degrees of freedom (DOF)

The associate editor coordinating the review of this manuscript and approving it for publication was Hiram Ponce<sup>(D)</sup>.



FIGURE 1. A number of different multilegged robots that use OpenSHC, from left to right: Gizmo, Zee, Bullet, MAX and Weaver.

hexapod robot platform called Bullet in the nature inspired mammalian and insectoid (sprawling-type) configuration is presented. Six legged robots have an advantage over bipeds and quadrupeds when it comes to statically stable locomotion on mild terrain, where the fast tripod gait has an energy efficiency advantage over other gaits [5]. In the mammalian configuration, the legs are below the body, reducing the support polygon while decreasing the power consumption required to support the body. Insectoid configuration places the legs to the side of the body, lowering the centre of mass, increasing locomotion workspace and stability. The different physical and locomotion specifications require unique parameters for control, something OpenSHC allows for easily. The experimental results provide unique insights into the novel parameter space of hardware changes to leg arrangement and configuration; and locomotion changes to step frequency, stride length and gait.

The design philosophy and history behind OpenSHC is presented in Section II. The kinematic algorithms in the controller is summarised in Section III. Section IV provides an overview of OpenSHC with details of each sub-component of the system. Section V describes the study of locomotion efficiency with respect to popular leg configurations in literature, while Section VI describes the mechanical specifications of Bullet. Experiments are explained in Section VII with results shown in Section VIII and then discussed in Section IX. Section X concludes the paper and provides areas of focus for the future of OpenSHC.

#### **II. DESIGN PHILOSOPHY**

The design philosophy of OpenSHC was to create a modular controller for research and development on different simulated and real hardware robots without having to redevelop the controller to make each robot walk.

A variety of open-source multilegged robot projects exist, each focusing on different challenges and applications. Projects such as OpenRoACH [6], Open Dynamic Robot Initiative [7] and Oncilla [8] have focused on developing a complete robot platform with mechanical and electrical hardware designs, and control software that is tightly coupled to the hardware specifications. OpenSHC on the other hand, is developed to be applicable to many different legged robots including those in simulation. We also differ from the Phoenix [9] robot control software used to control Trossen Robotics PhantomX hexapods [10], through full ROS [11] integration, the de facto robotics middleware used by the research community and industry. OpenSHC is a controller that is not linked to a particular hardware design. That is, OpenSHC allows robot morphology to be iteratively tested in simulation first, before being deployed onto a real system.

Through its full ROS integration, OpenSHC has the ability to easily interface with simulated robots in Gazebo, the ROS compatible 3D simulation engine, using its Gazebo control interface. This allows testing of various features and design parameters in designing a robot without having to go through hardware based iterations. This is done through the use of a Universal Robot Description Format (URDF) file which describes the robot model's links, joints, simulated actuators and IMU. The URDF file for the robot platform Bullet, used in the case study in Section V is provided as part of the suite of OpenSHC open source software packages. Using OpenSHC to command the joints of a robot model, a user can test custom robot morphologies, gaits, stance positions and various other parameters to optimise a robot design and test it in a full physics simulated environment. In development of the hexapod robot Bruce, used in the DARPA SubT Challenge [12], initial analysis of the actuator requirements was undertaken using OpenSHC and an initial design model of Bruce with simulated actuators in Gazebo. This analysis involved commanding the initial Bruce model to run at a variety of velocities and gaits, with a variety of payload weights and recording the simulated actuator torque requirements. Those requirements were used to inform the final design of Bruce including morphology and choice of actuator. Legged robots such as MAX [13], a 2.25 m 18 DOF hexapod; Weaver [14] and Bullet [2], both 30 DOF hexapod robots; and Magneto [15], a quadruped with 3 DOF actuated limbs and 3 DOF compliant magnetic feet have also been extensively simulated using OpenSHC to develop new applications and functionality (Fig. 2).

The following list highlights some of the key features of OpenSHC:

- Fully configurable for a variety of platform designs with differing physical characteristics, including up to 8 legs each with up to 6 DOF per leg.
- Dynamically switchable gait options with ability to design custom gaits.
- User defined body clearance, step clearance and step frequency.
- Manual body posing in 6 DOF.
- Manual leg manipulation (legipulation) for up to two legs simultaneously with toggle of manipulation control of either tip position in Cartesian space or direct control of joint positions (3 DOF legs only).



FIGURE 2. OpenSHC used for simulating multilegged robots in Gazebo, from left to right: Bullet, MAX, Weaver and Magneto.

- Startup direct mode to move foot tip positions linearly from initial position to default walking stance positions.
- Startup sequence mode with full chain of startup/ shutdown sequences to start from a 'packed' state and generate a sequence to stand up off the ground into its default walking stance; and similarly able to shutdown and transition back to a packed state.
- Cruise control mode to set robot velocity as a constant predefined input velocity or set to the current input velocity.
- Auto navigation mode when interfaced with high level navigation stack.
- Optional admittance control with dynamic leg stiffness to ensure leg contact with ground and offer moderately rough terrain walking ability.
- Optional IMU body compensation to keep body horizontally level at all times, using IMU data.
- Optional inclination compensation which strives to keep body centre of gravity over the estimated centroid of the support polygon whilst walking on inclined planes.
- Optional bespoke automatic body posing system to pose each robot leg cyclically as defined by auto-pose parameters.

A variety of research projects conducted in our lab has contributed and benefited from the controller. Novel functionality developed on particular robot platforms are generalised and incrementally added into the controller so other platforms can benefit from the work. Different robot platforms such as MAX, Weaver and Magneto mentioned earlier, have influenced the design and functionality of the controller. The simple command interface for sending velocity control to the system enabled higher level autonomy to be integrated and tested onto hardware easily. Research in probing for brittle terrain [2], steep terrain ascent [16], adapting robot pose for confined spaces [17], augmented telepresence for remote inspection [18] and autonomous adaptation of locomotion parameters [19]–[21] have all built upon the OpenSHC framework.

#### **III. KINEMATIC MODEL**

In OpenSHC, the robot is represented as a kinematic model within the controller using the Denavit-Hartenberg (DH) parameters [4], [22]. This allows for the joint angles and end effector locations to be easily transformed for forward

kinematics (FK) and inverse kinematics (IK) calculations. The robot's body frame is represented as pose  $(o_b x_b y_b z_b)$ , consisting of the body orientation  $(o_{roll}, o_{pitch}, o_{yaw})$  and displacement in (x, y, z). This robot frame provides the odometry of the robot to a reference map frame  $(o_m x_m y_m z_m)$  which is fixed to the world. The coordinate frames are right-handed with *x* forward, *y* left and *z* up. Each leg has its own leg frame  $(o_1 x_1 y_1 z_1)$ , which is the static offset from the robot's body frame origin to the centre of rotation of the first joint of that leg, when moving outward from the robot centre. The leg frame is used to transform the desired robot body movement to the corresponding leg motions via FK and IK.

The leg number convention used in OpenSHC (shown in Fig. 3) follows a clockwise sequence from the front right leg as '1' [23]. This convention facilitates arbitrary body designs such as elongated or axis-symmetric circular bodies [24]. Joint and link names are bio-inspired from insect morphology [25]. The links are named coxa, femur, tibia and tarsus and the joints immediately before the link is given the same name as the link for easy reference (E.g. tibia joint is between the femur and tibia links). Where there are compound joints with multiple degrees of freedom, the individual degrees of freedom is used as a subscript with the relevant link and joint names (E.g.  $\cos a_{yaw}$ ,  $\cos a_{roll}$ ).

#### A. FORWARD KINEMATICS

The specified DH parameters  $\theta_i$ ,  $d_i$ ,  $a_i$  and  $\alpha_i$  are the rotation around *z*, translation along *z*, translation along *x* and rotation around *x*, respectively [26]. The representation of the combined homogeneous transform of the parameters becomes:

$$H_{i+1}^{t} = Rot_{z,\theta_{i}} \cdot Trans_{z,d_{i}} \cdot Trans_{x,a_{i}} \cdot Rot_{x,\alpha_{i}}$$

$$= \begin{bmatrix} c_{\theta_{i}} - s_{\theta_{i}}c_{\alpha_{i}} & s_{\theta_{i}}s_{\alpha_{i}} & a_{i}c_{\theta_{i}} \\ s_{\theta_{i}} & c_{\theta_{i}}c_{\alpha_{i}} & -c_{\theta_{i}}s_{\alpha_{i}} & a_{i}s_{\theta_{i}} \\ 0 & s_{\alpha_{i}} & c_{\alpha_{i}} & d_{i} \\ 0 & 0 & 0 & 1 \end{bmatrix}$$

$$(1)$$

where  $c_x$  and  $s_x$  denote  $\cos(x)$  and  $\sin(x)$  respectively. Details on the transformation matrices are provided in Appendix A. For further information about FK, please see [26].

In the notation  $H_{i+1}^i$ , the superscript denotes the reference frame  $(o_i x_i y_i z_i)$  and the subscript indicates the transformed frame  $(o_{i+1} x_{i+1} y_{i+1} z_{i+1})$ . Using Fig. 3 as a 30 DOF hexapod example, for a leg of the robot the transform from the leg



FIGURE 3. The kinematic structure of the hexapod Bullet, showing leg numbering, link and joint naming conventions used in OpenSHC.

frame  $(o_1x_1y_1z_1)$  to the end effector  $(o_ex_ey_ez_e)$  is given by:

$$H_e^1 = H_2^1(q_1) \cdot H_3^2(q_2) \cdot H_4^3(q_3) \cdot H_5^4(q_4) \cdot H_e^5(q_5)$$
(2)

where  $q_1, q_2, q_3, q_4$  and  $q_5$  are the joint angles for the coxayaw, coxaroll, femur, tibia and tarsus joints respectively.

#### **B. INVERSE KINEMATICS**

The IK for calculating the desired change in joint angles ( $\Delta \theta$ ) for a given incremental change in end effector position  $(\Delta \vec{s})$ is calculated using the Jacobian matrix J and the Levenberg-Marquardt method, also known as the damped least squares method. Full derivation is provided in Appendix B (based on [27]), with the resultant equation to solve given by:

$$\Delta \theta = J^T \left( J J^T + \lambda^2 I \right)^{-1} \Delta \vec{\mathbf{s}} = \mathbf{Z} \Delta \vec{\mathbf{s}}$$
(3)

where  $\lambda > 0 \in \mathbb{R}$  and  $\mathbf{Z} = J^T (JJ^T + \lambda^2 I)^{-1}$ .

For robots with redundant DOFs, joint position/velocity limit avoidance (JLA) strategies are used to increase the safe operation of the joints [28]. A full-time kinematic optimisation method of JLA is used to keep the joints away from joint position and velocity limits as much as possible. The cost function to minimise uses the p-norm of a vector to approximate the focusing on the joint which is farthest from its centre, given by:

$$\Phi\left(q\right) = \left(\sum_{i=1}^{n} \left| K_{ii} \frac{q_i - q_{c_i}}{\Delta q_i} \right|^p \right)^{\frac{1}{p}}$$
(4)

where  $q_{c_i}$  is the centre of the joint range  $\Delta q_i$  for joint *i*, and K is the matrix for the weights of the joint importance.

Alg	Algorithm 1 Locomotion Workspace Search			
1:	for $h \in [H_{min}, H_{max}]$ do			
2:	for $\alpha \in [0, 2\pi]$ do			
3:	$endSearch \leftarrow false,$			
4:	$d_{max} \leftarrow 0, P_{target} \leftarrow P_0$			
5:	while !endSearch do			
6:	$P_{desired} \leftarrow P_{target}$			
7:	$j_{solved} \leftarrow solveIK(P_{desired})$			
8:	for $j \in j_{solved}$ do			
9:	<b>if</b> <i>!withinJointLimits(j)</i> <b>then</b>			
10:	$endSearch \leftarrow true$			
11:	else			
12:	$P \leftarrow solveFK(j)$			
13:	$move(foot \rightarrow P)$			
14:	if $ P - P_{desired}  > \Delta_P$ then			

For the optimised solution to the cost function, **v** is solved by:

endSearch  $\leftarrow$  true

 $d_{max} \leftarrow |P_{target} - P_0|$ 

 $P_{target} \leftarrow increment(P_{target}, \alpha, h)$ 

end if

end if

end for

end while

end for

 $r_{\alpha,h} = d_{max}$ 

$$\mathbf{v} = -\nabla \Phi. \tag{5}$$

Equation (5) can be optimised for a combined cost function of position and velocity limits. Combining (3) and (5) of the joint limit costing function, the solution for the required change in joint positions  $\mathbf{q}_{\theta}$  is given by:

$$\mathbf{q}_{\theta} = \mathbf{Z}\Delta \vec{\mathbf{s}} + (I - \mathbf{Z}J)\,\mathbf{v}.\tag{6}$$

The weighting for the preference of joint position limit avoidance or joint velocity limit avoidance is able to be customised within OpenSHC.

#### C. LEG WORKSPACE

15:

16: 17:

18:

19:

20:

21:

22.

23:

24: end for

The locomotion workspace is pre-computed at system initialisation and used to define the workspace available for each leg while walking. It is centred around the stance foot tip position of each leg and is calculated iteratively through moving the desired foot tip along different directions in the xy plane until the kinematics fail. This process is given in Algorithm 1 which is executed for each foot in the robot model. Here, h is the search height, incremented by a predefined  $\Delta_h$  height step between minimum and maximum search heights  $H_{min}$  and  $H_{max}$ . The symbol  $\alpha$  is the search bearing angle, incremented by a predefined  $\Delta_{\alpha}$  bearing step. The maximum distance the foot tip travels from its origin position is  $d_{max}$ .  $P_0$ ,  $P_{target}$ ,  $P_{desired}$  and P are the origin tip position, target tip position, desired tip position and actual tip position respectively. The maximum error tolerance between actual and desired



FIGURE 4. Final result of the locomotion workspace search for four different legged robots visualised in rviz. Top down view: the black outlines represent the robot model, the blue lines show the 3D leg tip workspace limited by morphology and joint limits while the red lines show the restricted planar non-overlapping locomotion sub-space of the workspace referred to as the *walkspace*.

tip positions is denoted by  $\Delta_P$ . The function *solveIK*() is the inverse kinematics solver producing joint positions for a given foot tip position and *solveFK*() is the forward kinematics solver producing a foot tip position for a given set of joint angles. *jsolved* is the set of joint angles for a given leg's joints for a desired tip position  $P_{desired}$ . The workspace radius for each foot at a given search bearing  $\alpha$  and height *h* is  $r_{\alpha,h}$ .

The result is a volumetric workspace, with the different *z*-axis height slices forming a polyhedron for each leg. The top-down view of these polyhedra for four different robots is illustrated in Fig. 4. A single restricted workspace polygon at the desired body height is called the *walkspace*. The walkspace is constrained to be symmetrical, with the minimum walkspace from all the legs selected so that the polygon is the same for each leg. The combination of nonoverlapping workspaces and JLA are used to prevent selfcollisions without a dedicated collision check module. The walkspace is also used to calculate and limit the stride length for a given desired input body velocity. The desired body velocity is passed to the robot for a given step frequency, with the corresponding stride length calculated to achieve the desired body velocity. The maximum linear body velocity is thus limited by the walkspace of the legs. The body velocity calculated by OpenSHC is given by:

$$\nu_{body} = \frac{l_s \times f_s}{\beta} \tag{7}$$

where  $\beta$  is the duty factor defined as  $\beta = T_{stance}/T_{stride}$ , the time in stance phase, and the total time of stance and swing phase for the stride time;  $l_s$  the stride length; and  $f_s$  the step frequency. Note that both  $l_s$  and  $f_s$  are not fully independent, as  $f_s$  is limited by the maximum joint velocity, which is also affected by the required distance to move from the stride length. In reality, the actual velocity is lower due to slippage and other disturbances. However, this is not accounted for in OpenSHC. Therefore, for accurate robot odometry, an external tracking solution is required. More examples of leg workspaces are presented in Section VI-A and Fig. 18.



FIGURE 5. The hexapod Weaver in (a) rviz kinematic model, (b) Gazebo simulation and (c) hardware.

## **IV. SYSTEM ARCHITECTURE**

The high-level controller consists of multiple modules, and is wrapped as a C++ node within the Robot Operating System (ROS) [11] framework. The utilisation of rostopics and the ROS parameter server allows for a common interface to work with various inputs (for control and sensors) and outputs (position control of joints), and easy dynamic parameter customisation. With the integration of Gazebo [29] with ROS, robot algorithms can be tested in simulation before being deployed onto hardware. Fig. 5 shows the robot Weaver as a pure kinematic model in rviz, a simulation model in Gazebo and the real robot hardware.

The modules of OpenSHC can be customised via configuration parameters for each robot platform, with additional functionality modules configurable to be enabled or disabled depending on the scenario. For example, for inclined, uneven terrain, the additional functionality of the different pose generators (Section IV-C) and admittance controller (Section IV-D) can be enabled to increase stability. Fig. 6 shows the simplified structure of OpenSHC with its core functionality. The coloured lines show control flow in typical use. The subsequent figures breakout each of the three main controllers in to greater detail. These are the walk controller in Fig. 7, the pose controller in Fig. 11 and the robot controller in Fig. 12.

The control inputs received by OpenSHC passes the desired body velocity and pose velocity to the walk controller and pose controller whose control flows are shown in green and red respectively. The tip pose generator combines the tip trajectory with the body pose and passes it to the robot controller. This takes the resultant tip pose and calculates the desired joint angles. The joint controller calls the IK and FK systems and performs a final check to constrain the output joint positions to safe limits. The output joint states of Open-SHC are either sent to the motor controller which controls the servomotors, or to the control interface of a simulation engine such as Gazebo. Typical control flow through the robot controller is shown in blue.

#### A. CONTROL INPUTS

OpenSHC receives desired body velocities (forward, lateral and angular) and body posing velocities (6 DOF) via an



FIGURE 6. OpenSHC's hierarchical control architecture shown in a simplified view containing typically used components only. The coloured arrows show the flow of information in each controller during typical use. The green, red and blue lines show control flow through the walk controller, the pose controller and the robot controller respectively.

external source. The source can be an operator via a gamepad, tablet, computer, or from an autonomous navigation stack. The interface takes in a linear velocity vector along the x, y, z axes and an angular velocity vector about the x, y, z axes, using the ROS *geometry\_msgs::Twist* message as transport. For the walk controller, the system uses the linear x and y velocities and angular z velocity, with linear z velocity and angular velocities about x and y being ignored. For the pose controller, all six velocities are used to pose the robot's body.

The user is able to select which external source to use, allowing auto navigation or cruise control mode, along with manual control. Auto navigation requires commands from an autonomous navigation stack that provides obstacles avoidance functionality, while cruise control sets the robot's



**FIGURE 7.** Detailed diagram of the walk controller with typical control flow shown in green.

velocity constant in the desired velocity generator within the walk controller.

## **B. WALK CONTROLLER**

The walk controller (Fig. 7) generates the required foot tip trajectories and timings from the desired body velocity. Within the walk controller, the gait configuration parameters are converted into timings for each leg in the gait timing generator (Section IV-B1) which are then transformed to foot tip trajectories in the tip trajectory generator (Section IV-B2). The desired velocity generator fits the body velocity to the walkspace while the stride vector generator calculates the required stride length using (7) to achieve the desired velocity. The following subsections describe the gait timing generator and the tip trajectory generator, two major components of the walk controller, in detail.

## 1) GAIT TIMING GENERATOR

Working along with the stride vector generator to achieve the desired velocity, the gait timing generator takes in predefined parameters defining the different gaits; such as wave, amble, ripple, tripod and the dynamic bipod gaits [5], [30]. The parameters define the ratio between the stance and swing periods of each gait cycle and the phase offset between each leg executing the gait cycle, as shown in Fig. 8. The timing of these gait cycles and the desired state of the leg is sent to the tip trajectory generator which generates either a swing or stance trajectory based on the current state of the leg in the gait cycle.

#### 2) TIP TRAJECTORY GENERATOR

The tip trajectory generator calculates the foot tip trajectories for the specified gait pattern and stride vector. It consists of three 4<sup>th</sup> order Bézier curves to generate foot positions and velocities over the time period of the step cycle. Two Bézier curves control the foot trajectory across the primary



FIGURE 8. Timing for predefined gaits for a hexapod robot.

(first half) and secondary (second half) swing period, and the third Bézier curve controls the stance period. The stance Bézier curve generates foot velocities rather than positions using the derivative of the position Bézier curves used in the swing period. This is to ensure that on-ground foot velocities are guaranteed to match those required to achieve the desired body velocity. Each Bézier curve is defined by 5 control points and the following equations:

$$B(t) = s^4 P_0 + 4ts^3 P_1 + 6s^2 t^2 P_2 + 4st^3 P_3 + t^4 P_4 \quad (8)$$

where s = 1 - t and  $t \in [0, 1]$ . The derivative of (8) is given by:

$$B'(t) = 4s^{3}(P_{1} - P_{0}) + 12s^{2}t(P_{2} - P_{1}) + 12t^{2}s(P_{3} - P_{2}) + 4t^{3}(P_{4} - P_{3}).$$
(9)

The control points are placed to maximise smoothness throughout the step cycle whilst ensuring desired swing characteristics, such as step clearance and liftoff/touchdown placement, as well as required ground velocity during stance.



FIGURE 9. The control points for the three Bézier curves that form the foot tip trajectory.



FIGURE 10. The foot tip trajectory as defined by the three Bézier curves and associated control points, ensuring desired stride length, step clearance, step width and swing depth.

Position continuity is ensured between step cycle periods, as shown in Fig. 9, by setting the 'end' control points for each Bézier curve at the same location. Similarly, control point placement strategies are employed for the remaining control points to ensure velocity and acceleration continuity, producing trajectories which are at least  $C^1$  smooth and preferably  $C^2$  smooth within the swing characteristic constraints. The desired characteristics of the swing and stance periods have precedence over trajectory smoothness. Open-SHC ensures that the foot trajectories will always adhere to three requirements:

- The foot trajectory during swing is position controlled to ensure it achieves the defined liftoff and touchdown positions at the start and end respectively of the primary and secondary swing Bézier curves.
- The foot trajectory during swing is position controlled to ensure it achieves the highest point at the desired step clearance directly above a defined 'default' foot position, i.e. the position of the foot with zero body velocity.
- The foot trajectory during stance is velocity controlled to ensure the required foot velocity is attained as per the requirements of the desired body velocity.

The resultant foot trajectory from the control points is illustrated in Fig. 10.

# 3) MANUAL LEG MANIPULATION

A 'freegait' module, within the walk controller but not shown in Fig. 7 due to being decoupled from the typical control flow, allows for leg tip poses to be controlled outside of the cyclic gait. The desired tip pose of any leg can be controlled by either specifying the velocity of the leg tip or the actual pose of the leg tip in reference to the leg frame for basic manual leg manipulation tasks (legipulation). The freegait tip pose overrides the tip trajectory generator output to the tip pose generator. Achieving the desired tip pose is not guaranteed, subject to IK being able to solve for valid joint angles for the DOFs available.

## C. POSE CONTROLLER

The pose controller (Fig. 11) generates the required tip pose to achieve the input manual body pose and sensor based body pose for safe operation. Changing the body pose from the default position and orientation allows the robot to shift its centre of mass for increased stability on rough terrain. Additionally, sensors mounted onto the robot's body with limited field of view can be oriented towards an area of interest by changing the body pose. The manual pose generator receives the desired body posing velocities from the control input and alters the robot's body pose within the maximum limits.

The following subsections give details of five main components of the pose controller: IMU pose generator, inclination pose generator, current body pose generator, startup/shutdown sequence generator and the tip pose generator.

## 1) IMU POSE GENERATOR

The IMU pose generator provides feedback to correct the robot's body position. The IMU pose generator makes use of the inertial measurement unit (IMU) to ensure the robot body does not tend towards an unstable position. As the body experiences changes in roll and pitch, it is posed in opposing roll and pitch axes to correctly align the robot frame *z*-axis parallel to the direction of the gravity vector to keep the body level. This posing controller uses a PID loop to ensure the posing is stable and remains within predefined limits.

## 2) INCLINATION POSE GENERATOR

During locomotion in heavily inclined terrain, if the body is oriented such that correction from posing in roll/pitch cannot stabilise it, then the inclination pose generator poses the body laterally in the *xy* plane of the robot frame according to the IMU orientation input. The goal of the system is to centre the vertically projected robot centre of mass into the centre of the support polygon of the robot's load bearing legs.

## 3) CURRENT BODY POSE GENERATOR

The current body pose generator adds together the output of several separate body posing systems whilst ensuring legs do not exceed the maximum translation and rotation limits of the body. In addition to the two main body posing sub-systems



**FIGURE 11.** Detailed diagram of the pose controller showing typical control flow in red.

in Sections IV-C1 and IV-C2, the body pose generator takes input from: the walk plane pose generator which aligns the body to a *walkplane* estimate; the auto pose generator which is an automatic user defined cyclical body pose generator; and the tip aligned pose generator which shifts the body to allow 3 DOF legs to achieve desired tip orientations upon touchdown. Each body posing sub-system transforms the current body pose generator output  $P_{body}$  as summarised by:

$$P_{body} = H_{ali}H_{AI}H_{inc}H_{man}P_{walk}$$
(10)

where  $H_{man}$ ,  $H_{inc}$  and  $H_{ali}$  are the transformation matrices calculated by manual pose, inclination pose and tip align pose respectively.  $H_{AI}$  is either IMU pose or auto pose, as only one can be active at a time without interference.  $P_{walk}$  is the pose parallel to the walkplane, which is the plane of best fit from the foot tip contact points. The reference frame of  $P_{walk}$  is  $P_{default}$ , the default pose given by zero translation and identity orientation, except for the body clearance. The output poses of sub-systems that have similar functionality and incompatible with each other (such as IMU pose or auto pose) are removed from the output pose to ensure the system is not overcompensating for disturbances.

#### 4) STARTUP/SHUTDOWN SEQUENCE GENERATOR

The sequence generator turns a list of joint angles for each joint to achieve sequentially, to moving the robot joints to the desired angles. This provides smooth motions from any initial leg position to the leg positions for stance. The sequence generator also allows for the robot to move to a packed state where the legs are tucked in for easy transport. Intermediate steps allows for overlapping packed legs to be unpacked safely without leg collisions.

## 5) TIP POSE GENERATOR

The tip pose generator combines the desired body pose output of the body pose generator and the desired tip pose output of the tip trajectory generator to create the resultant tip pose for each leg of the robot. Each leg tip pose  $P_{leg}$  is referenced from the default body pose frame  $P_{default}$ , the same reference frame used for  $P_{walk}$ . Thus, the inverse transform is used to transform the leg tip pose  $P_{leg}^{default}$  to  $P_{leg}^{body}$ .

The result of each leg's tip pose generator enables the robot to enact desired foot tip trajectories for desired walking characteristics whilst simultaneously posing the body as desired. This module also ensures tip poses do not exceed the given workspaces for each leg.

#### D. ROBOT CONTROLLER

The robot controller (Fig. 12) takes the desired poses and timings of the foot tip and treats it independently for each leg via the leg controllers. It is able to detect foot tip touchdown to stop the leg motors over-torquing and to model the walk plane of the robot. the following subsections describe two of the main controllers within the leg controllers; admittance controller and joint controller.

## 1) ADMITTANCE CONTROLLER

The admittance-based leg controller is used for rough terrain locomotion and is based on the research in [14]. A virtual elastic element is added to the legs to compensate for non-planar foot placements. Joint torques are mapped to the force at the foot tip using a Jacobian  $J_e$  as follows:

$$\begin{bmatrix} F_e^2(t) \\ M_e^2(t) \end{bmatrix} = (J_e(q_1, q_2, q_3, q_4, q_5)^T)^{-1} \cdot M_q$$
(11)

where  $M_q = [M_1 \ M_2 \ M_3 \ M_4 \ M_5]^T$  is the vector of joint torques and  $q_1, q_2, q_3, q_4, q_5$  are the five joint angles corresponding to joints coxayaw, coxa<sub>roll</sub> femur, tibia and tarsus.  $F_e^2 = [F_x \ F_y \ F_z]^T$  and  $M_e^2 = [M_x \ M_y \ M_z]^T$  represents the 3D force and torque vectors at the foot tip. The admittance controller takes the force at the foot tip as an input and outputs a displacement along the *z* axis. As shown in Fig. 13, a virtual mass  $m_{virt}$ , virtual stiffness  $c_{virt}$  and virtual damping element  $b_{virt}$  defines the dynamic behaviour along the *z* axis. This second order system is represented by:

$$-F_z = m_{virt} \dot{\Delta z_r} + b_{virt} \dot{\Delta z_r} + c_{virt} \Delta z_r.$$
(12)

The exerted force and the foot position is adapted by this virtual second order mechanical system by reacting to measured foot force. This adapted foot position is  $z_d = z_r - \Delta z_r$ . The displacements  $\Delta x_r$  and  $\Delta y_r$  are set to zero in order to restrict motion in the *x* and *y* directions during foot contact.

#### 2) JOINT CONTROLLER

The joint controller handles interaction with the forward and inverse kinematics systems for every joint in each leg to achieve a desired foot tip pose. It provides leg state and model information for the associated joints for the FK and IK



**FIGURE 12.** Detailed diagram of the robot controller showing typical control flow in blue.



**FIGURE 13.** Admittance control modelled as a mass spring damper mechanical system.

systems to solve (2) and (6) respectively. It also restricts any joints to ensure position and velocity limits are not exceeded. The controller acts with the FK and IK systems to produce joint commands which, once sent to the motors, will execute the desired pose of the associated foot tip as close as possible whilst adhering to joint limits.

To convert the desired joint states to movement of the motors, motor interface packages are required. The integration with ROS allows for modularity with different servo motors in the robot hardware. For simulation, a Gazebo model of the robot is controlled through the output of OpenSHC using the ROS joint\_state\_controller [31].

## V. CASE STUDY: OPTIMAL LEGGED LOCOMOTION

The versatility of OpenSHC enables rapid customisation of the locomotion parameters to enable different hardware configurations to move. The underlying kinematics and trajectory algorithms allow various parameters to be optimised.



**IEEE**Access

FIGURE 14. Hexapod robot Bullet walking in insectoid (top) and mammalian (bottom) configurations.

Coupled with 3D printing for rapid hardware iterations, OpenSHC is used for hardware optimisation of the locomotion parameter space for a hexapod robot named Bullet, shown in Fig. 14.

The mammalian and insectoid configurations are favoured in nature for different animals with different number of legs. Predominately, four legged animals have a mammalian configuration while six legged animals have an insectoid (arachnid/sprawling-type) configuration. This is reflected in robotic research as well; mammalian quadruped robots such as ANYbotics' ANYmal [3], Boston Dynamics' Spot (successor of BigDog [32]), IIT's HyQ [33] and MIT's Cheetah 3 [34] have sophisticated controllers capable of dynamic locomotion that are fast and agile. Insectoid hexapod robots, such as MAX [13], LAURON V [1], DLR-Crawler [35] and SILO6 [36] have been specifically designed for the rough terrain found in disaster zones, demining areas and complex environments. For effective locomotion, the mammalian configuration uses a roll-pitch-pitch configuration for its leg joints. The pitch joints propel the body forwards, while the roll joints are used for stability and to aid steering. For insectoid, a yaw-pitch-pitch configuration is used, with the yaw joints mainly used to propel the body and steer. The pitch joints are required to bear the body weight, even when stationary, consuming more energy compared to mammalian configuration.

Research in unique locomotion principles have resulted in designs of insectoid quadrupeds and mammalian



FIGURE 15. Bullet in insectoid and mammalian configuration in kinematic arrangements A and B.

hexapods. The TITAN XIII [37], aimed to use the wider range of leg motion of an insectoid configuration on a quadruped to achieve energy efficient dynamic walking. MRWALLSPECT-III [38], a climbing robot also used this design to allow for a greater workspace with a low mass. SILO6, a hexapod robot, has been analysed for energy efficient configurations in [39]. Theoretical and experimental analysis conducted on flat terrain showed that the mammalian configuration in a hexapod produced a lower power consumption overall, compared with insectoid. However, SILO6 could not switch between insectoid and mammalian configurations on-the-fly and required manual intervention to mechanically change the leg mounting for this to be possible. Theoretical analysis of mammalian multilegged robots is presented in [40], with a set of performance indices described for optimisation.

Previous works in optimising the locomotion parameter space have focused on tuning parameters for a single physical embodiment of the robot. Both learning [41] and nonlearning [20] based methods have been used to optimise for energy efficient locomotion. The focus of this work is to expand the locomotion parameter search into the unexplored area of optimising locomotion efficiency for different physical configurations of an over-actuated hexapod robot, along with the traditional locomotion parameters (step frequency, stride length and gait type). While the study is not an exhaustive search of this space, the hardware-based experimental results provide insight into this unique space.

Bullet was designed to overcome locomotion workspace limitations of Weaver [14] and is capable of switching between insectoid and mammalian leg configurations. In addition to statically stable gaits, we also evaluate the performance of Bullet when using the dynamic gait bipod-B introduced in [30]. The bipod-B gait for hexapods provides a

TABLE 1.	Hardware Specifications of Bullet. A and B denotes the tw	o
kinematio	c arrangements presented.	

Туре	Description
General	Mass (without battery): 8.41 kg (A), 9.47 kg (B)
	Mass (with battery): 10.07 kg (A), 11.13 kg (B)
	Dimensions (Body): $L_B = 500 \text{ mm x } W_B = 280 \text{ mm}$
Servomotors	$30 \times \text{Dynamixel MX-106}$
Power supply	Motors: 7-cell LiPo battery (25.9 V, 5000 mAh)
	Computer: 4-cell LiPo battery (14.8 V, 3300 mAh)
Computer	Intel i7 NUC PC (16 GB RAM) running ROS on Ubuntu
-	16.04
Sensors	IMU (Microstrain GX5 - 100 Hz)
	Arduino based power monitor (90 Hz)

fast and energy optimal gait when foot tips have low traction. With experimental results, we show that the mammalian configuration outperforms the insectoid configuration in energy efficiency at the expense of stability at low step frequencies and maximum body velocity.

## **VI. ROBOT PLATFORM**

Bullet is a versatile 30 DOF hexapod robot designed for traversing rough terrain. Bullet's body length has been extended compared to Weaver's, with legs offset further and perpendicular to the body to allow for the legs to be folded beside the body for a mammalian configuration without limiting the workspace with self-collisions. Fig. 15 illustrates Bullet in insectoid and mammalian configurations. Bullet is capable of un-tethered remote operation with on-board batteries, computer and sensors (described in Table 1).

#### A. LEG KINEMATIC ARRANGEMENT AND WORKSPACE

Two different leg kinematic arrangements were designed through analysis of workspace and joint limits. The original kinematic arrangement (arrangement A), first introduced



Mammalian leg configuration

Insectoid leg configuration





**FIGURE 17.** Bullet's leg joint configuration in both mammalian (left) and insectoid (right) configurations in kinematic arrangement B.

in [20], consists of a yaw-roll-pitch-pitch-pitch kinematic arrangement. This arrangement is compared with a new pitch-yaw-pitch-pitch kinematic arrangement (arrangement B) with extended link lengths. These joints are named coxavaw, coxaroll, femur, tibia and tarsus respectively for the former and coxapitch, coxayaw, femur, tibia and tarsus respectively for the latter. Arrangement B increases total leg length from 340 mm to 466 mm and improves joint ranges by up to 20° through reduction of self collision. The extended link lengths were set as the minimum length required for the greater joint limits, to lessen the impact of flex and increased motor torques. The change improves the 3D workspace, increases the joint limits and allows the robot to operate inverted (mirrored about the xy plane). Fig. 16 and 17 illustrates these leg kinematic arrangements in mammalian and insectoid configuration.

Fig. 18 illustrates the different workspaces for each leg configuration and kinematic arrangement with the values

#### TABLE 2. Calculated Workspace of Different Configurations and Arrangements.

Config.	Arrange.	Area (m <sup>2</sup> )	Stride Len. (m)
Mammalian	А	0.049	0.207
	В	0.063	0.210
Insectoid	А	0.046	0.287
	В	0.060	0.348



**FIGURE 18.** Calculated workspace for the different configurations and arrangements. The rectangle represents the robot body with the robot coordinate frame's origin at its centre. The red dots represent the foot tip locations.

outlined in Table 2. The workspace area is calculated per leg and the stride length is taken as the forwards (robot's x-axis) distance of the workspace assuming forward motion.

## **B. LEG CONFIGURATION**

Bullet's unique five DOF legs and body shape allows for selfactuated switching of the leg configuration between mammalian and insectoid without external intervention, different to [39]. In the insectoid configuration, five DOFs are available to position and orientate the leg tarsus. To solve for the foot tip position of the over-actuated leg, an additional constraint is used to solve the inverse kinematics problem. The foot tip orientation is constrained to return to the original stance orientation for consistent foot tip touchdown. During the swing phase, the IK solver is able to utilise all joints for motion, following the algorithms outlined in Section III-B. This is different to [20], where the tarsus and coxa<sub>roll</sub> joints were constrained by the inclination of the robot.

To transform Bullet into the mammalian configuration, the  $coxa_{yaw}$  joints are commanded and locked to fixed positions for kinematic arrangement A. The direction of the leg orientation (either forwards or backwards) is controlled through the  $coxa_{yaw}$  joint rotation. For arrangement A, the resulting four DOF leg contains the  $coxa_{roll}$  joint for direction and the three pitch (femur, tibia and tarsus) joints for propulsion. In arrangement B, the  $coxa_{yaw}$  joint is limited, but not locked. This arrangement utilises the  $coxa_{yaw}$  joint for direction (by bending the leg inwards or outwards) and  $coxa_{pitch}$ , femur, tibia and tarsus joints for propulsion. Similar to the insectoid configuration, the leg is over-actuated, with the foot tip orientation constrained during touch down.

For mammalian configuration in quadrupeds, research has found marginal differences in stability depending on the elbows and knees pointing forwards or backwards [42]. In hexapods, the leg orientation of choice is front leg forwards, and middle and rear legs backwards [39], [43]. This orientation was implemented on kinematic arrangement A without success on Bullet. This was due to stability issues when walking in tripod gait caused by Bullet's centre of mass and the limited calculated leg workspace. The workspace was limited due to the femur and tibia joints reaching their limits. Through analysis of the workspace and stability while walking, the legs backwards configuration was found to be the most stable for walking while maximising the workspace that borders with the adjacent legs. Due to the legs bending backwards, the rear two sets of legs were required to be further behind the body to keep the centre of mass within the support polygon, causing the asymmetrical leg positions for the mammalian configuration as shown in Fig. 18. Leg arrangement B follows the same orientation for consistency.

#### C. PARAMETER SPACE

The parameter search space is defined by physical and locomotion parameters. The physical parameters encapsulate the leg configuration (mammalian and insectoid) and kinematic arrangement (A: yaw-roll-pitch-pitch-pitch or B: pitch-yawpitch-pitch). From these physical parameters, the locomotion parameters of: gait type, step frequency and stride length are optimised. These parameters are summarised as:

$$\Pi = C \times A \times G \times L_s \times F_s \tag{13}$$

where the sets  $C = \{Mammalian, Insectoid\}, A = \{A, B\}, G = \{tripod, bipod\}, L_s = \{60, 75, 90, 100\}$ as a percentage of maximum stride length and  $F_s = \{0.4, 0.6, 0.8, 1.0, 1.2, 1.4, 1.6, 1.8, 2.0, 2.2\}$  in Hz. The search for the optimal combinations were limited to motions that were safe for the robot, with some combinations not investigated.

#### **VII. EXPERIMENTS**

The performance of Bullet was tested across flat terrain to evaluate the effectiveness of the different configurations.

#### A. Performance Metrics

The power consumption and joint torques were compared in stance (standing stationary) and while walking on flat terrain. The dimensionless energetic cost of transport (CoT) is a performance metric used to compare locomotion of animals [44] and also wheeled and legged robots [14], [37], [45].

#### TABLE 3. Controller Parameters for Insectoid and Mammalian.

Config.	Arrange.	Stride height (m)	Body height (m)
Mammalian	А	0.12	$H_m = 0.30$
	В	0.08	$H'_{m} = 0.30$
Insectoid	А	0.15	$H_i = 0.20$
	В	0.10	$H'_{i} = 0.25$

The overall cost of transport  $(\overline{CoT})$  over a travelled distance is given by:

$$\overline{CoT} = \frac{\frac{1}{n} \sum_{i=1}^{n} U_i I_i}{mg \frac{\Delta x}{\Delta t}}$$
(14)

where U is the power supply voltage (V), I is the instantaneous power supply current draw (A), n is the total number of data points, m is the mass (kg), g is the gravitational acceleration ( $ms^{-2}$ ) and  $\Delta t$  is the time needed in seconds to travel distance  $\Delta x$  (m). Herein, CoT refers to the overall  $\overline{CoT}$ .

The *CoT* depends on the velocity of the robot [44], where consumption is high at low and high speeds, with a local minimum value at a particular speed. The desired body velocity for the controller is governed by (7), assuming perfect conditions, ignoring slippage, robot model tolerances, leg link flex and motor errors. Thus, a higher desired velocity can be achieved through increasing either the step frequency or stride length parameter. To account for non-perfect conditions, the body velocity was tracked externally for *CoT* calculations.

To evaluate the effectiveness of the different configurations, the tripod gait, known for its stability and speed, was compared between mammalian and insectoid at different speeds. Other statically stable gaits such as amble, wave or ripple were not tested as previous works have shown tripod is optimal on flat terrain. The insectoid bipod-B gait [30], was also compared. OpenSHC allows for parameters, listed in Table 3, to be set. These values were heuristically tuned to optimise for energy efficiency using the rules listed in [39].

#### **B. EXPERIMENTAL SETUP**

The power consumption and energy efficiency for the different leg configurations were tested in four scenarios. These scenarios break down the different terms that contribute to power consumption and are:

- SA Power consumed in stance elevated in the air (robot lifted up, no ground contact),
- WA Power consumed walking elevated in the air (robot lifted up, no ground contact),
- SG Power consumed in stance on the ground, and
- WG Power consumed walking on the ground.

These scenarios follow the experiments set out in [39] where: scenario SA (stance air) provides the power consumed by the motor power circuitry; scenario WA (walk air) provides the power consumed for leg movements; scenario SG (stance ground) for supporting the robot's body weight; and scenario WG (walk ground) for overall power consumption. The robot

**TABLE 4.** Average Power Consumption (W).

Config. A	Arrang	e. Gait	Stance air	Walk air	Stance ground
Mammalian	А	Tripod	27.7	51.0	48.6
	В	Tripod	30.4	47.1	46.7
Insectoid	Α	Tripod	30.4	61.9	58.3
	Α	Bipod	30.4	72.3	58.3
	В	Tripod	31.1	62.9	56.0

was elevated on a stand with legs in mid-air for scenarios SA and WA. The robot was placed on the ground and walked once in place to measure the power consumption when stationary in scenario SG. Bullet was analysed traversing flat ground in scenario WG. A flat, slightly inclined 3 m straight track was marked for the robot. For the calculation of CoT in (14), power consumption (P = UI) was logged with a power monitor at 90 Hz and velocity v was measured with a total station (Leica TS12) at approximately 4 Hz. The CoT is based on the power draw of the motors, including mechanical energy, heat dissipation and friction, while the computer and sensors power consumption is assumed to be constant across the different configurations and arrangements, and are powered on a separate power supply. Bullet was tested using a tether to an external power supply and control computer, with the internal batteries and computer on-board to simulate the real weight of the robot. The power supply was set at 25 V and the control computer is an i7 laptop with 8 GB of RAM.

#### **VIII. RESULTS**

The metric used to compare scenarios SA, WA and SG is power consumption with results outlined in Table 4, while CoT is used for scenario WG. The power consumed in stance for both kinematic arrangements of mammalian is lower than their corresponding insectoid. This validates the theory of less torque on the motors with the legs underneath as the force vectors act through the motor shaft. The difference in power between the air and ground shows the additional power required to support the robot's weight. While arrangement A consumes less power than B in stance while in the air, on the ground arrangement B consumes less. This suggests that the joints in arrangement B require less torque to keep the body up. The walking in air comparison is for step frequency 1 Hz at 100% stride length. The results show the insectoid bipod consuming the highest amount of energy for leg motion, while in mammalian, arrangement B uses less power than arrangement A.

An analysis of arrangement A mammalian configuration at different step frequency and stride length compared to the CoT is shown in Fig. 19. The stride length, given as a percentage of the available locomotion workspace, shows that a higher length for a given step frequency is more energy efficient. The minimum value at 1 Hz shows the optimal step frequency and stride length for energy efficient locomotion. An increase of step frequency above 1 Hz increases the power usage greatly. At a step frequency of 1.4 Hz, the foot tips slipped considerably, causing the tracked velocity to be lower



**FIGURE 19.** *CoT* for arrangement A mammalian tripod gait at various step frequencies and stride lengths.

(a)



FIGURE 20. CoT for insectoid bipod and tripod, and mammalian tripod gait at various step frequencies at 100% stride length for arrangement A.

than that of step frequency 1.2 Hz; even though the desired body velocity is greater.

The insectoid configuration was analysed with the tripod and bipod gait in arrangement A. A stride length of 100% was used in tests as results in mammalian showed a stride length of 100% is the most efficient across all step frequencies tested. Fig. 20 compares the *CoT* and average velocity as the step frequency increases. Similar to mammalian configuration, tripod gait has a minimum *CoT*, occurring at 1.8 Hz. An increase of step frequency above 1.8 Hz increases the *CoT*. The bipod gait does not exhibit this minimum, instead having diminished reductions for step frequencies about 2 Hz. The bipod gait performed equally or better than the tripod gait at all step frequencies. For step frequencies above 2 Hz, both insectoid gaits cause considerable clamping of the joint angles by OpenSHC. This caused body instability and the foot

(b)



**FIGURE 21.** *CoT* for mammalian and insectoid tripod gait at various step frequencies and stride lengths for arrangement B. Insectoid stride length 60% was limited to be the same metric length as mammalian 100% stride length.

tip to no longer track the trajectory. Step frequencies above 2.2 Hz were not tested to prevent hardware damage.

For arrangement B, tests were conducted to compare the mammalian and insectoid configuration given the same workspace across various step frequencies. The mammalian configuration was tested at 100% stride length. This was compared with insectoid at 60.27% of its maximum workspace to have the same metric stride length as mammalian. Fig. 21 outlines the *CoT* and average velocity as the step frequency increases. Insectoid at 100% stride length was included for comparison. At low step frequencies, the mammalian configuration was observed to have body oscillation and foot tip slip, resulting in the high CoT. At step frequency 2 Hz and above, where the body oscillations were minimal, the mammalian configuration has comparable or better performance than both insectoid stride lengths. Given the same stride length, mammalian configuration consistently has a lower CoT than insectoid across step frequencies above 2 Hz (Fig. 21a). However, the actual robot velocity is consistently lower than insectoid (Fig. 21b). The local minimum value for CoT occurs at step frequency 3 Hz for both configurations, suggesting it is the step frequency and stride length that affects the local minimum CoT and not the leg configuration.

The comparison of 100% stride length between the two configurations show the advantages and disadvantages of each. The insectoid configuration can achieve a higher maximum velocity at lower step frequencies and is also more stable. However, it causes the joints to reach limits at a lower frequency (observed at 2 Hz compared to mammalian at 3.5 Hz). The drop in velocity in insectoid at high step frequencies is caused by joint velocity clamping while executing the desired controller trajectory. As the weight of the robot is constant across the different trials, the *CoT* is effectively the relative power required to traverse a set distance, and can be



**FIGURE 22.** Velocity  $(rads^{-1})$  profile of the joints in the front right leg for arrangement A. The solid blue line denotes a step frequency of 1 Hz at 100% stride length. The dotted red line denotes the maximum step frequency at 100% stride length for each configuration (1.4 Hz, 2.2 Hz and 2.2 Hz respectively). The *x*-axis denotes two periods of the step frequency at 1 Hz.



**FIGURE 23.** Load profile (dimensionless) of the joints in the front right leg for arrangement A. The solid blue line denotes a step frequency of 1 Hz at 100% stride length. The dotted red line denotes the maximum step frequency at 100% stride length for each configuration (1.4 Hz, 2.2 Hz and 2.2 Hz respectively). The *x*-axis denotes two periods of the step frequency at 1 Hz.

directly used to compare energy efficiency of the different configurations.

#### **IX. DISCUSSION**

The use of an over-actuated leg to achieve switching between mammalian and insectoid configuration was explored for the first time. To achieve the mammalian configuration, the  $coxa_{yaw}$  joint motors were commanded to set positions, with torque holding those positions. This was reflected by the small velocity and load on the joint illustrated in Fig. 22a and 23a respectively. Comparing the load on the joints across the different gaits (Fig. 23 blue line), the

mammalian tripod gait had lower torques on all the joints. The results for the stance power consumption experiments (scenarios SA and SG) confirmed the theory in [46] and also the experimental data in [39] with the mammalian configuration using less energy than insectoid when standing.

Arrangement A mammalian configuration consumed less power than insectoid in stance (SG) and leg movement (WG), but was not the most efficient in locomotion due to the smaller locomotion workspace and joint angle limits. The robot in insectoid configuration was capable of covering larger distances in the same number of steps, negating the higher power required to support its weight and for locomotion. With the joint limit improvements in arrangement B, the mammalian configuration achieved lower *CoT* than insectoid. The mammalian configuration was optimal if the least amount of power to traverse an area is required. Arrangement B mammalian did not reach joint velocity limits until higher step frequencies as the pitch-yaw-pitch-pitch-pitch kinematic arrangement affords four pitch joints to propel the robot forwards.

Analysis of the insectoid tripod and bipod gait showed a lower CoT for the latter. Although the bipod gait consumed more power, it was able to walk faster than tripod at the same step frequency. This supports the hardware experiments conducted in [30]. On average, the bipod gait was 30% faster than the equivalent tripod gait, similar to the 25% increase reported in [30]. Between the two arrangements and across different gaits, arrangement A bipod had the lowest CoT. The results showed improved locomotion efficiency is achievable with OpenSHC, providing direction for future works in the absence of sophisticated dynamic controllers.

The versatility of OpenSHC was highlighted on experiments conducted on a versatile hexapod capable of switching between different configurations. We show that this switching ability has advantages over traditional fixed configuration robots in autonomous navigation of terrain. When traversing mild terrains, where stability and speed is of less importance, the mammalian configuration consumes less power. For terrain or scenarios where stability or speed is required (such as rough terrain or speed critical tasks), the robot can morph into insectoid configuration. This ability to select the optimal configuration provides the robot an advantage on locomotion efficiency for real world scenarios. Having an over-actuated leg design allows for the additional ability to selectively use particular motors for locomotion and to switch between the different configurations. While a single DOF is unused in the mammalian configuration, setting the motor to a fixed position results in the robot using less power overall. The results highlight the advantages of a versatile hexapod running OpenSHC, capable of selecting between configurations and step frequencies, based on energy efficiency (mammalian) or stability and speed (insectoid) requirements.

## **X. CONCLUSION**

This paper presented OpenSHC, a versatile controller capable of generating smooth trajectories for quasi-static

troller can be configured for new robot platforms with various physical characteristics. OpenSHC provides the building blocks to extend the capabilities of multilegged robot platforms through a modular hierarchical architecture. By providing 'out of the box' functionality for locomotion in mild to rough terrain, OpenSHC allows researchers to focus on higher level areas such as autonomy or application specific motions that would increase the versatility of legged robots. With additional sensors, robots would be capable of traversing in more complex and confined environments, something that quasi-static multilegged platforms would be ideal for. Another area of research that OpenSHC supports is in leg manipulation. For robot platforms with greater than 3 DOF legs, OpenSHC allows for manual position and orientation control of the foot tip, allowing attachments such as grippers to manipulate the environment. Additionally, with its seamless integration with rviz and Gazebo simulation environments, OpenSHC provides a way for researchers to design, test and optimise new legged robots in simulation and make informed choices about hardware based on application requirements.

legged robots. With many customisable parameters, the con-

## **APPENDIX A FORWARD KINEMATICS DERIVATION**

The DH parameters  $\theta_i$ ,  $d_i$ ,  $a_i$  and  $\alpha_i$  used in (1) from Section III-A are represented as transformation matrices using the shorthand  $c_x$  and  $s_x$  to denote  $\cos(x)$  and  $\sin(x)$ respectively.

The transformation matrix for the DH parameter  $\theta_i$  is given by:

$$Rot_{z,\theta_i} = \begin{bmatrix} c_{\theta_i} & -s_{\theta_i} & 0 & 0\\ s_{\theta_i} & c_{\theta_i} & 0 & 0\\ 0 & 0 & 1 & 0\\ 0 & 0 & 0 & 1 \end{bmatrix}$$
(15)

The transformation matrix for the DH parameter  $d_i$  is given by:

$$Trans_{z,d_i} = \begin{bmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & d_i \\ 0 & 0 & 0 & 1 \end{bmatrix}$$
(16)

The transformation matrix for the DH parameter  $a_i$  is given by:

$$Trans_{x,a_i} = \begin{bmatrix} 1 & 0 & 0 & a_i \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 \end{bmatrix}$$
(17)

The transformation matrix for the DH parameter  $\alpha_i$  is given by:

$$Rot_{x,\alpha_i} = \begin{bmatrix} 1 & 0 & 0 & 0 \\ 0 & c_{\alpha_i} & -s_{\alpha_i} & 0 \\ 0 & s_{\alpha_i} & c_{\alpha_i} & 0 \\ 0 & 0 & 0 & 1 \end{bmatrix}$$
(18)

#### **APPENDIX B INVERSE KINEMATICS DERIVATION**

The IK for the change in joint angles for a given desired end effector position is calculated using the Jacobian. The Jacobian matrix J is defined as:

$$J(\theta) = \left(\frac{\partial s_i}{\partial \theta_j}\right)_{i,j} \tag{19}$$

where  $s_i$  is the *end effector* of the link which is affected by joint  $\theta_j$ . The linear velocity component of the Jacobian can be calculated by:

$$\frac{\partial s_i}{\partial \theta_i} = \mathbf{v}_j \times \left(\mathbf{s}_i - \mathbf{p}_j\right) \tag{20}$$

where  $v_j$  is the unit vector along the current axis of rotation and  $p_j$  is the position of the *j*<sup>th</sup> joint. For incremental iterations of the end effector, a change in the end effector can be approximated by:

$$\Delta \vec{\mathbf{s}} \approx J \Delta \theta. \tag{21}$$

Thus, using the Levenberg-Marquardt method, also known as the damped least squares method,  $\Delta\theta$  can be solved by:

$$\Delta \theta \approx J^T \left( J J^T + \lambda^2 I \right)^{-1} \Delta \vec{\mathbf{s}}$$
 (22)

where  $\lambda > 0 \in \mathbb{R}$ . Using this approximation and compact notation, we get:

$$\Delta \theta = J^T \left( J J^T + \lambda^2 I \right)^{-1} \Delta \vec{\mathbf{s}} = \mathbf{Z} \Delta \vec{\mathbf{s}}$$
(23)

where  $\mathbf{Z} = J^T (JJ^T + \lambda^2 I)^{-1}$ .

The cost function to minimise for JLA in OpenSHC is represented by:

$$\Phi(q) = \sum_{i=1}^{n} \left[ \frac{q_i - q_{c_i}}{\Delta q_i} \right]^2$$
(24)

where  $q_{c_i}$  is the centre of the joint range  $\Delta q_i$  for joint *i*.

## ACKNOWLEDGMENT

The authors would like to thank O. Jayasinghe, M. Bucolo, J. Brett, I. Kalhara, E. Tennakoon, T. Molnar, B. Wilson, T. Lowe, and D. Rytz for their support during this project.

#### REFERENCES

- A. Roennau, G. Heppner, M. Nowicki, and R. Dillmann, "LAURON V: A versatile six-legged walking robot with advanced maneuverability," in *Proc. IEEE/ASME Int. Conf. Adv. Intell. Mechatronics*, Jul. 2014, pp. 82–87.
- [2] E. Tennakoon, T. Peynot, J. Roberts, and N. Kottege, "Probe-before-step walking strategy for multi-legged robots on terrain with risk of collapse," in *Proc. IEEE Int. Conf. Robot. Autom.*, May/Aug. 2020 pp. 5530–5536.
- [3] M. Hutter, C. Gehring, D. Jud, A. Lauber, C. D. Bellicoso, V. Tsounis, J. Hwangbo, K. Bodie, P. Fankhauser, M. Bloesch, R. Diethelm, S. Bachmann, A. Melzer, and M. Hoepflinger, "ANYmal—A highly mobile and dynamic quadrupedal robot," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Oct. 2016, pp. 38–44.
- [4] R. Hartenberg and J. Danavit, *Kinematic Synthesis of Linkages*. New York, NY, USA: McGraw-Hill, 1964.
- [5] N. Kottege, C. Parkinson, P. Moghadam, A. Elfes, and S. P. N. Singh, "Energetics-informed hexapod gait transitions across terrains," in *Proc. IEEE Int. Conf. Robot. Autom.*, May 2015, pp. 5140–5147.

- [6] L. Wang, Y. Yang, G. Correa, K. Karydis, and R. S. Fearing, "Open-RoACH: A durable open-source hexapedal platform with onboard robot operating system (ROS)," in *Proc. Int. Conf. Robot. Autom.*, May 2019, pp. 9466–9472.
- [7] F. Grimminger, A. Meduri, M. Khadiv, J. Viereck, M. Wüthrich, M. Naveau, V. Berenz, S. Heim, F. Widmaier, T. Flayols, J. Fiene, A. Badri-Spröwitz, and L. Righetti, "An open torque-controlled modular robot architecture for legged locomotion research," *IEEE Robot. Autom. Lett.*, vol. 5, no. 2, pp. 3650–3657, Apr. 2020.
- [8] A. T. Spröwitz, A. Tuleu, M. Ajallooeian, M. Vespignani, R. Möckel, P. Eckert, M. D'Haene, J. Degrave, A. Nordmann, B. Schrauwen, J. Steil, and A. J. Ijspeert, "Oncilla robot: A versatile opensource quadruped research robot with compliant pantograph legs," *Frontiers Robot. AI*, vol. 5, p. 67, Jun. 2018. [Online]. Available: https://www.frontiersin.org/article/10.3389/frobt.2018.00067
- [9] J. Janssen, K. Halvorsen, K. Eckhardt, M. E. Ferguson, and B. Porter. (2018). *Phantom Phoenix*. [Online]. Available: https://github. com/KurtE/Phantom\_Phoenix
- [10] Trossen Robotics. (2020). PhantomX Hexapod Robot Kits. [Online]. Available: https://www.trossenrobotics.com/Quadruped-Robot-Hexapod-Robot-Kits.aspx
- [11] M. Quigley, K. Conley, B. Gerkey, J. Faust, T. Foote, J. Leibs, R. Wheeler, and A. Y. Ng, "ROS: An open-source robot operating system," in *Proc. IEEE ICRA Workshop Open Source Softw.*, vol. 3, May 2009, p. 5.
- [12] R. Steindl, T. Molnar, F. Talbot, N. Hudson, B. Tam, S. Murrell, and N. Kottege, "Bruce—Design and development of a dynamic hexapod robot," in *Proc. Australas. Conf. Robot. Automat.*, 2020.
- [13] A. Elfes, R. Steindl, F. Talbot, F. Kendoul, P. Sikka, T. Lowe, N. Kottege, M. Bjelonic, R. Dungavell, T. Bandyopadhyay, M. Hoerger, B. Tam, and D. Rytz, "The multilegged autonomous eXplorer (MAX)," in *Proc. IEEE Int. Conf. Robot. Autom.*, May 2017, pp. 1050–1057.
- [14] M. Bjelonic, N. Kottege, and P. Beckerle, "Proprioceptive control of an over-actuated hexapod robot in unstructured terrain," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Oct. 2016, pp. 2042–2049.
- [15] T. Bandyopadhyay, R. Steindl, F. Talbot, N. Kottege, R. Dungavell, B. Wood, J. Barker, K. Hoehn, and A. Elfes, "Magneto: A versatile multilimbed inspection robot," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Oct. 2018, pp. 2253–2260.
- [16] T. Molnar, R. Steindl, N. Kottege, F. Talbot, and A. Elfes, "Steep terrain ascension controller for hexapod robots," in *Proc. Australas. Conf. Robot. Autom.*, 2017.
- [17] R. Buchanan, T. Bandyopadhyay, M. Bjelonic, L. Wellhausen, M. Hutter, and N. Kottege, "Walking posture adaptation for legged robot navigation in confined spaces," *IEEE Robot. Autom. Lett.*, vol. 4, no. 2, pp. 2148–2155, Apr. 2019.
- [18] B. Tam, N. Kottege, and B. Kusy, "Augmented telepresence for remote inspection with legged robots," in *Proc. Australas. Conf. Robot. Autom.*, 2017.
- [19] T. Homberger, M. Bjelonic, N. Kottege, and P. V. K. Borges, "Terraindependent motion adaptation for hexapod robots," in *Proc. Int. Symp. Experim. Robot.*, 2016, pp. 92–102.
- [20] M. Bjelonic, T. Homberger, N. Kottege, P. Borges, M. Chli, and P. Beckerle, "Autonomous navigation of hexapod robots with vision-based controller adaptation," in *Proc. IEEE Int. Conf. Robot. Autom.*, May 2017, pp. 5561–5568.
- [21] M. Bjelonic, N. Kottege, T. Homberger, P. Borges, P. Beckerle, and M. Chli, "Weaver: Hexapod robot for autonomous navigation on unstructured terrain," *J. Field Robot.*, vol. 35, no. 7, pp. 1063–1079, Oct. 2018.
- [22] M. W. Spong, S. Hutchinson, and M. Vidyasagar, *Robot Modeling and Control*. Hoboken, NJ, USA: Wiley, 2006.
- [23] D. Belter and P. Skrzypczyński, "A biologically inspired approach to feasible gait learning for a hexapod robot," *Int. J. Appl. Math. Comput. Sci.*, vol. 20, no. 1, pp. 69–84, Mar. 2010.
- [24] D. C. Clemente, Gait Optimization for Multi-Legged Walking Robots, With Application to a Lunar Hexapod. California, CA, USA: Stanford Univ., 2011.
- [25] G. M. Nelson, R. D. Quinn, R. J. Bachmann, W. C. Flannigan, R. E. Ritzmann, and J. T. Watson, "Design and simulation of a cockroachlike hexapod robot," in *Proc. IEEE Int. Conf. Robot. Autom.*, vol. 2, Apr. 1997, pp. 1106–1111.

- [26] K. J. Waldron and J. Schmiedeler, "Kinematics," in *Springer Handbook of Robotics*, B. Siciliano and O. Khatib, Eds. Cham, Switzerland: Springer, 2016, pp. 11–36.
- [27] S. R. Buss, "Introduction to inverse kinematics with jacobian transpose, pseudoinverse and damped least squares methods," Dept. Math., Univ. California, San Diego, CA, USA, Tech. Rep., 2004. [Online]. Available: https://math.ucsd.edu/~sbuss/ResearchWeb/ikmethods/iksurvey.pdf
- [28] F. Fahimi, "Redundant manipulators," in Autonomous Robots: Modeling, Path Planning, and Control, F. Fahimi, Ed. Boston, MA, USA: Springer, 2009, pp. 1–36.
- [29] N. Koenig and A. Howard, "Design and use paradigms for Gazebo, an open-source multi-robot simulator," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, vol. 3, Oct. 2004, pp. 2149–2154.
- [30] P. Ramdya, R. Thandiackal, R. Cherney, T. Asselborn, R. Benton, A. J. Ijspeert, and D. Floreano, "Climbing favours the tripod gait over alternative faster insect gaits," *Nature Commun.*, vol. 8, no. 1, p. 14494, Apr. 2017.
- [31] S. Chitta, E. Marder-Eppstein, W. Meeussen, V. Pradeep, A. R. Tsouroukdissian, J. Bohren, D. Coleman, B. Magyar, G. Raiola, M. Lüdtke, and E. F. Perdomo, "Ros\_control: A generic and simple control framework for ROS," *J. Open Source Softw.*, vol. 2, no. 20, p. 456, 2017.
- [32] M. Raibert, K. Blankespoor, G. Nelson, and R. Playter, "BigDog, the rough-terrain quadruped robot," in *IFAC Proceedings Volumes*, vol. 41, no. 2. Amsterdam, The Netherlands: Elsevier, 2008, pp. 10822–10825.
- [33] C. Semini, V. Barasuol, J. Goldsmith, M. Frigerio, M. Focchi, Y. Gao, and D. G. Caldwell, "Design of the hydraulically actuated, torque-controlled quadruped robot HyQ2Max," *IEEE/ASME Trans. Mechatronics*, vol. 22, no. 2, pp. 635–646, Apr. 2017.
- [34] G. Bledt, M. J. Powell, B. Katz, J. Di Carlo, P. M. Wensing, and S. Kim, "MIT cheetah 3: Design and control of a robust, dynamic quadruped robot," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Oct. 2018, pp. 2245–2252.
- [35] M. Gorner, T. Wimbock, A. Baumann, M. Fuchs, T. Bahls, M. Grebenstein, C. Borst, J. Butterfass, and G. Hirzinger, "The DLR-crawler: A testbed for actively compliant hexapod walking based on the fingers of DLRhand II," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Sep. 2008, pp. 1525–1531.
- [36] P. Gonzalez de Santos, J. A. Cobano, E. Garcia, J. Estremera, and M. A. Armada, "A six-legged robot-based system for humanitarian demining missions," *Mechatronics*, vol. 17, no. 8, pp. 417–430, Oct. 2007.
- [37] S. Kitano, S. Hirose, A. Horigome, and G. Endo, "TITAN-XIII: Sprawling-type quadruped robot with ability of fast and energy-efficient walking," *Robomech J.*, vol. 3, no. 1, Dec. 2016.
- [38] T. Kang, H. Kim, T. Son, and H. Choi, "Design of quadruped walking and climbing robot," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, vol. 1, Oct. 2003, pp. 619–624.
- [39] D. Sanz-Merodio, E. Garcia, and P. G. de Santos, "Analyzing energyefficient configurations in hexapod robots for demining applications," *Ind. Robot Int. J.*, vol. 39, no. 4, pp. 357–364, 2012.
- [40] M. F. Silva and J. A. T. Machado, "Kinematic and dynamic performance analysis of artificial legged systems," *Robotica*, vol. 26, no. 1, pp. 19–39, 2008.
- [41] E. Heijmink, A. Radulescu, B. Ponton, V. Barasuol, D. G. Caldwell, and C. Semini, "Learning optimal gait parameters and impedance profiles for legged locomotion," in *Proc. IEEE-RAS Int. Conf. Humanoid Robot.*, Nov. 2017, pp. 339–346.
- [42] S. Meek, J. Kim, and M. Anderson, "Stability of a trotting quadruped robot with passive, underactuated legs," in *Proc. IEEE Int. Conf. Robot. Autom.*, May 2008, pp. 347–351.
- [43] B. Jin, L.-J. Zhao, J.-L. Zhang, C. Chen, and S. Hu, "Design of the control system for a hexapod walking robot," in *Proc. 2nd Int. Conf. Digit. Manuf. Autom.*, Aug. 2011, pp. 401–404.
- [44] V. A. Tucker, "The energetic cost of moving about: Walking and running are extremely inefficient forms of locomotion. much greater efficiency is achieved by birds, fish and bicyclists," *Amer. Scientist*, vol. 63, no. 4, pp. 413–419, 1975.
- [45] S. Seok, A. Wang, M. Yee Chuah, D. Otten, J. Lang, and S. Kim, "Design principles for highly efficient quadrupeds and implementation on the MIT cheetah robot," in *Proc. IEEE Int. Conf. Robot. Autom.*, May 2013, pp. 3307–3312.
- [46] D. J. Todd, Walking Machines: An Introduction to Legged Robots. Boston, MA, USA: Springer, 1985.



**BENJAMIN TAM** (Student Member, IEEE) received the B.E. degree in mechatronic engineering from The University of Queensland, Brisbane, QLD, Australia, in 2016. He is currently pursuing the Ph.D. degree with the School of Information Technology and Electrical Engineering, The University of Queensland while affiliated to the Robotics and Autonomous Systems Group, CSIRO, Pullenvale, QLD, Australia.

His current research interests include legged locomotion, mobile manipulation, and robot navigation.



**FLETCHER TALBOT** (Member, IEEE) received the B.E. degree in mechatronic engineering from The University of Queensland, Brisbane, QLD, Australia, in 2013.

He joined CSIRO, Pullenvale, QLD, Australia, in 2014. He is currently an Engineer with the Robotics and Autonomous Systems Group, CSIRO. He specialises in control software for multi-legged robots. He is also the Lead Developer of the software stack for the high-level con-

trol of CSIRO multi-legged robots, including the multi-legged autonomous eXplorer (MAX). He also participates as a key member of the CSIRO team competing in the DARPA Subterranean Robotics Challenge. Outside of legged robotic locomotion, his research interests include HMI, robot navigation, and behavioural robotics.



**RYAN STEINDL** (Member, IEEE) received the B.E. degree in mechatronics engineering from the Queensland University of Technology, Brisbane, QLD, Australia, in 2012.

He joined CSIRO, Pullenvale, QLD, Australia, in early 2013. He is currently a Senior Mechatronics Engineer with the Robotics and Autonomous Systems Group, CSIRO. He works at CSIRO revolves around robotic system design with a focus on legged locomotion. His current research

revolves around the development of a dynamic legged robot platform that has competed in the DARPA SubT challenge. His robotic design achievements include MAX, CSIRO's Ultralight Legged Robot, which demonstrates a paradigm shift in design, construction and control; Amazon robot, a single man deployable, legged platform that explored the Amazon rainforest floor of the Mamiraua reserve in Brazil; Gizmo, a small legged inspection robot intended to explore roof cavities; Zee, a medium sized inspection robot intended to conduct final build inspection inside aircraft wings; and Magneto, a magnetic foot climbing robot capable of traversing any incline including full inversions.



**ALBERTO ELFES** (Member, IEEE) received the B.Eng. degree in electronics engineering and the M.Sc. degree in computer science from the Aeronautics Institute of Technology (ITA), Brazil, in 1975 and 1980, respectively, and the Ph.D. degree in electrical and computer engineering from Carnegie Mellon University (CMU), Pittsburgh, PA, USA, in 1989, where he developed the Occupancy Grid framework for robot mapping and navigation.

He joined CSIRO, Pullenvale, QLD, Australia, in 2011. He was a Chief Research Scientist with the Robotics and Autonomous Systems Group, CSIRO. Before joining CSIRO, he held a number of senior positions, including the Director of the Brazilian Automation Institute and a Principal Researcher at the NASA's Jet Propulsion Laboratory (JPL). He has led projects in autonomous aerial, ocean, and ground robots for Earth science missions and planetary exploration. His research interests include 3D multimodal perception and legged robots for extreme terrain. He was a recipient of the Mercator Professorship Award of the German Research Foundation (DFG) and several NASA awards. He was also an Honorary Professor with The University of Queensland, Brisbane, QLD, Australia.



**NAVINDA KOTTEGE** (Senior Member, IEEE) received the B.Sc. degree in engineering physics from the University of Colombo, Sri Lanka, in 2003, and the Ph.D. (engineering) degree from The Australian National University (ANU), Canberra, ACT, Australia, in 2009, where he developed a relative localisation system for small autonomous underwater vehicles.

He joined CSIRO, Pullenvale, QLD, Australia, in 2009. He is currently a Senior Research Scien-

tist and a Research Team Leader with the Robotics and Autonomous Systems Group, CSIRO, where he leads legged robotics research. His research interests include perception and control for legged locomotion in unstructured terrain. He is also the Chair of IEEE Queensland joint chapter of Robotics and Automation and Control Systems Societies. He is also a Honorary Senior Lecturer with The University of Queensland and an Adjunct Associate Professor with the Queensland University of Technology.

. . .